Author's response to reviews

Title: Hemodynamic impact of isobaric levobupivacaine versus hyperbaric bupivacaine for subarachnoid anesthesia in patients aged 65 and older undergoing hip surgery

Authors:

ROSA HERRERA (35606rhc@gmail.com)
JOSE DE ANDRÉS (deandres_jos@gva.es)
LUIS ESTAÑ (estan@uv.es)
FRANCISCO MORALES-OLIVAS (morales@uv.es)
INOCENCIA MARTÍNEZ-MIR (martinez_ino@gva.es)
THORSTEN STEINFELDT (steinfeldt@gmx.de)

Version: 2 Date: 25 January 2014

Author's response to reviews: see over
Cover Letter

Name of Corresponding Author
Drª. Rosa Herrera
Anesthesia, Critical Care, and Multidisciplinary Pain Management Department
General University Hospital
Avda. Tres Cruces s/n
46014- Valencia (Spain)
Phone: (+) 34 96 1972138
Fax: (+) 34 96 1972303 E-mail: 35606rhc@gmail.com

BMC Anesthesiology

Re (Title of Manuscript)

Hemodynamic impact of isobaric levobupivacaine versus hyperbaric bupivacaine for spinal anesthesia in patients aged and older undergoing hip surgery.

On behalf of my co-authors, I am submitting the enclosed material for possible publication in BMC Anesthesiology. It has not been submitted for publication nor has it been published in whole or in part elsewhere. I attest to the fact that all authors listed on the title page have read the manuscript, attest to the validity and legitimacy of the data and its interpretation, and agree to its submission to BMC Anesthesiology.

Description of the changes made

1. AUTHOR’S CONTRIBUTIONS
RH conceived and designed of the study, acquired the data, performed the statistical analysis and drafted the manuscript. J.DA and IM participated in the design and coordination of the study and revised it critically. F-JM and LE helped to draft the manuscript. TS reviewed the study. All authors read and approved the final manuscript.

2. ACKNOWLEDGEMENTS: We sincerely thank Elena García and Davinia Oltra for their contribution to the design of randomization and the various negotiations
with the Spanish Drug Agency.

3. DECLARATION OF CONFLICTS OF INTEREST: ‘The authors declare that they have no competing interests (conflicts of interest, sources of financial support, corporate involvement patent holdings, etc.) with Abbott Laboratories. Copyright transfer and the signatures of all authors will be sent if requested prior to publication of accepted manuscripts.

Signature                        Date

Rosa Herrera

Valencia, January 24th, 2014